Author's response to reviews

Title: Poor birth weight recovery among low birth weight/preterm infants following hospital discharge in Kampala, Uganda

Authors:

Flavia B Namiiro (bnflaviah@gmail.com)
Jamiir Mugalu (mugalu89@yahoo.com)
Ryan M McAdams (mcadams@washington.edu)
Grace Ndeezi (gracendeezi@yahoo.com)

Version: 6 Date: 28 December 2011

Author's response to reviews:

28/12/11

Cover letter

Dear editor,

Thank you for looking at our work.

Below are the further formatting changes we had to make.

Major revisions

Author list: the list of authors in the manuscript should be written exactly as they are in the submission system, both in style and order. The preferred style is "First name Initial Last name" (e.g. Joe F Bloggs)

This is the order list and the names appear as; Flavia B Namiiro, Jamiir Mugalu, Ryan M McAdams, Grace Ndeezi

Minor revisions

It is important for the final layout of the manuscript that the figures are cropped as closely as possible to minimize white space around the image. This has been done

Typography: This has been checked

Thank you

Flavia B Namiiro